Author's response to reviews

Title: Experience with a new prosthetic anal sphincter in three coloproctological centres.

Authors:

Matthias Goos (matthias.goos@uniklinik-freiburg.de)
Ulrich Baumgartner (u.baumgartner@krankenhaus-emmendingen.de)
Mathias Löhnert (mathias.loehnert@klinikumbielefeld.de)
Oliver Thomusch (oliver.thomusch@uniklinik-freiburg.de)
Günther Ruf (guenter.ruf@uniklinik-freiburg.de)

Version: 2 Date: 21 December 2012

Author's response to reviews:

The present study is not a registered clinical trial. Therefore we delivered no TRN. The study reflects the experience of the three centers with a new medical product. The data collection was performed prospectively. The study was approved by the ethical committee. The new title and changes in method section reflect this.